Author's response to reviews

Title: Cancer incidence and adverse pregnancy outcome in registered nurses potentially exposed to antineoplastic drugs

Authors:

Pamela A Ratner (pam.ratner@ubc.ca)
John J Spinelli (jspinelli@bccrc.ca)
Kris Beking (kris.beking@vch.ca)
Maria Lorenzi (mlorenzi@bccrc.ca)
Yat Chow (yatc@interchange.ubc.ca)
Kay Teschke (kay.teschke@ubc.ca)
Nhu D Le (nle@bccrc.ca)
Richard P Gallagher (rgallagher@bccrc.ca)
Helen Dimich-Ward (helen.ward@vch.ca)

Version: 7 Date: 15 September 2010

Author's response to reviews:

Figure 1: The figure is referred to as 'Figure 1' throughout the manuscript.

The figure is cropped as closely as possible to minimise white space around the image.